\documentclass[11pt]{article}

\usepackage{fullpage,times,graphicx}

\frenchspacing

\long\def\comment#1{}

\begin{document}

\noindent
Name \rule{3in}{0.01in}\\[0.5in]
Email \rule{3in}{0.01in}

\begin{center}
\Large Reading Quiz for Sheather Chapter 5
\end{center}

~\\[3em]

\begin{enumerate}

\item Fill in the missing information in the following Analysis of Variance table, using the blanks below the table:\\[2em]


{\def\arraystretch{2}
\begin{tabular}{lllll}
\hline
Source of & Degrees of & Sums of squares & Mean square & F \\
variation & freedom (df) & (SS) & (MS) & \\
\hline
Regression & (a)  & $SSreg$ & $SSreg/p$ &  
$F = \mbox{(d)} $\\
Residual & (b)  & $RSS$ & (c) & \\
Total & $n-1$ & $SST$ & & \\
\hline
\end{tabular}
}

\begin{enumerate}

\item \rule{3in}{.01in}\rule{0in}{.75in}  % $p$

\item \rule{3in}{.01in}\rule{0in}{.75in}  % $n-p-1$

\item \rule{3in}{.01in}\rule{0in}{.75in}  % $RSS/(n-p-1)$

\item \rule{3in}{.01in}\rule{0in}{.75in}  % \frac{SSreg/p}{RSS/(n-p-1)}

\end{enumerate}

\end{enumerate}

\end{document}
